Author's response to reviews

Title: Willingness to Pay for Physician Services at a Primary Contact in Ukraine: Results of a Contingent Valuation Study

Authors:

Andriy Danyliv (danyliv@gmail.com)
Milena Pavlova (m.pavlova@maastrichtuniversity.nl)
Irena Gryga (griga@ukma.kiev.ua)
Wim Groot (w.groot@maastrichtuniversity.nl)

Version: 3 Date: 20 July 2012

Author's response to reviews:

Dear Editor,

The following changes was made to the manuscript.

1) In the Methods section the statement was added:
"The study instruments and methodology were reviewed by the Institutional Review Board at the National University of Kyiv-Mohyla Academy and a waiver form the full ethical review was obtained."

Best regards,

Andriy Danyliv